Author's response to reviews

Title: Association between fat mass and obesity associated (FTO) gene rs9939609 A/T polymorphism and polycystic ovary syndrome: A systematic review and meta-analysis

Authors:

Ai Ling Liu (LiuAiLing917@outlook.com)
Hui Jun Xie (729094650@qq.com)
Hong Yan Xie (19785642678@qq.com)
Jun Liu (nhuliujun@qq.com)
Jie Yin (13309624@qq.com)
Jin Song Hu (522371193@qq.com)
Cui Ying Peng (pengcuiying2004@126.com)

Version: 3 Date: 08 Aug 2017

Author’s response to reviews:

Response letter

Dear reviewers or editors:

I would like to thank all the reviewers and editors for their constructive comments and suggestions to improve the quality of the manuscript. I have carefully addressed all reviewers’ and editors’ comments point by point.

My point-by-point responses accompanied my revised manuscript are listed as follows:

1. I amended the “Figure Legends” section. I rewrote the legend of Fig2,3 and 4, and put it at the end of manuscript. (Figure Legend section, page 17). And I moved the titles of each figures in Figure Files. It just includes the number of figures and images.

2. I amend the “Ethics approval and consent to participate” and “Availability of data and material” section. (Declarations section, page 10-11)
Please note: My manuscript is a meta-analysis and all data came from the cited references. I contacted with all corresponding authors and asked for the consent for citing their papers, they all agreed!

I did upload a signal, final and clean version at this stage.

We thank again to all the reviewers and editors for their thoughtful and thorough review.